Author's response to reviews

Title: Feasibility and acceptability of point of care HIV testing in community outreach and GUM drop-in services in the North West of England: A programmatic evaluation

Authors:

Peter MacPherson (p.macpherson@liv.ac.uk)
Anu Chawla (Anu.Chawla@rlbuht.nhs.uk)
Kathy Jones (Kathy.Jones@rlbuht.nhs.uk)
Emer Coffey (Emer.Coffey@liverpoolpct.nhs.uk)
Vida Spaine (vida@liverpool.ac.uk)
Ian Harrison (ian.Harrison@liverpoolpct.nhs.uk)
Pauline Jelliman (pauline.jelliman@liverpoolpct.nhs.uk)
Penelope Phillips-Howard (P.Phillips-Howard@ljmu.ac.uk)
Caryl Beynon (C.M.Beynon@ljmu.ac.uk)
Miriam Taegtmeyer (miriamt@liverpool.ac.uk)

Version: 5 Date: 27 May 2011

Author's response to reviews:

Major and minor formatting changes made as requested